Author's response to reviews

Title: Feasibility of three wearable sensors for 24 hour monitoring in middle-aged women

Authors:

Jennifer L Huberty (jhuberty@asu.edu)
Diane K Ehlers (dkehlers@asu.edu)
Jonathan Kurka (jkurka@asu.edu)
Barbara Ainsworth (barbara.ainsworth@asu.edu)
Matthew P Buman (matthew.Buman@asu.edu)

Version: 6 Date: 6 May 2015

Author's response to reviews: see over
May 6, 2015

To Whom It May Concern:

Please accept the revised manuscript entitled, “Feasibility of three wearable sensors for 24 hour monitoring in middle-aged women” to BMC Women’s Health. We have addressed all of the reviewer comments and appreciate the time to review the paper.

The following revisions were made:
   1. Formatting of title page changed to reflect BMC requirement
   2. Formatting of references changed to reflect BMC requirement (e.g., bold, italics)

We are excited for the opportunity to have our work published in BMC Women’s Health.

All authors have read and approved the final paper. There are no conflicts of interest. The work was supported and approved by the Institutional Review Board at Arizona State University.

We look forward to the opportunity to have our work published in BMC Women’s Health.

Sincerely,

Jennifer Huberty, PhD